
Allen, Pam, NMENV 

From: 
Sent: 
To: 
Subject: 

May 

From: Kliphuis, Trais, NMENV 

Maestas, Ricardo, NMENV 
Wednesday, December 17, 2014 4:18 PM 
Allen, Pam, NMENV 
FW: WIPP Information - For Call Today 

Sent: Thursday, May 01, 2014 2:28 PM 
To: Flynn, Ryan, NMENV; Kendall, Jeff, NMENV 
Cc: Winchester, Jim, NMENV; Tongate, Butch, NMENV; Schwender, Erika, NMENV; Blaine, Tom, NMENV; Skibitski, 
Thomas, NMENV; Kieling, John, NMENV; Holmes, Steve, NMENV; LucasKamat, Susan, NMENV; Turner, Jill, NMENV; 
Nelson, Morgan, NMENV; Ines Triay (triayin@fiu.edu); Maestas, Ricardo, NMENV; Smith, Coleman, NMENV 
Subject: FW: WIPP Information - For Call Today 

NMED notes in blue 

From: Oba Vincent [mailto:oba.vincent@cbfo.doe.gov] 
Sent: Thursday, May 01, 2014 10:59 AM 
To: Kliphuis, Trais, NMENV; 'peake.tom@epa.gov'; 'Edwards, Jonathan'; 'Walsh, Jonathan'; 'Perrin, Alan'; 'Kehrman, Bob -
RES'; 'Chavez, Rick - RES'; 'Stone.Nick@epa.gov'; Smith, Coleman, NMENV; 'brozowski.george@epa.gov'; 'Fraass, Ron'; 
'Hardy, Russell'; 'Veal.Lee@epamail.epa.gov'; 'Economy, Kathleen (Economy.Kathleen@epa.gov)'; 'Poppell, Sam W. 
(Poppell.Sam@epa.gov)'; Maestas, Ricardo, NMENV 
Cc: George Basabilvazo - WIPPNet; 'Reynolds, Tammy - NWP'; 'Harris, Alton - DOE EM'; Susan McCauslin; 'Joe Harvill 
(jharvill@portageinc.com)'; 'Kennedy, Scott - NWP'; 'Jones, Stewart - RES'; 'Oates, Berta - CTAC'; 
'schultheisz.daniel@epa.gov'; Philip Theisen - ORISE; Russ Patterson - WIPPNet; 'Kouba, Steve - WRES'; Roger Nelson -
WIPPNet; 'Bignell, Dale - CTAC'; Susan McCauslin - WIPPNet; 'Pace, Berry'; Anthony Stone - WIPPNet; J.R. Stroble -
WIPPNet 
Subject: WIPP Information - For Call Today 

Attached are the action items listing, filter performance data and Station A/B data. Pictures and the video from 
yesterday have been posted on the portal. 

Below is a summary of activities: (Please note, activities and dates are subject to change. Please verify the most 
current dates of any information provided) . 

• Underground remains in Filtration Mode utilizing Fan 860B. This afternoon, the site will switch the ventilation to 

Fan 860C. 

• Cleaning of the soot from the fire event in the Waste Handling Tower is continuing; Cleaning of the 3rd/4th floor 

is underway. 

• Weekly MSHA inspections of the AIS & SS were performed on 4/29. 

• The mine entry was completed yesterday. Two teams entered the mine. One team replaced the filter paper on 

CAM 151 (31 papers per cartridge), which went into alarm when the event occurred. The other team went to 

the waste face in Panel 7, Room 7 and took additional photos and videos. They also took radiological smears 

from Panel 7, Room 6. The visual observations and photos confirm that several MgO bags were destroyed; 

however the MgO remains in place. It could not be determined visually, whether a thermal or chemical reaction 

destroyed the MgO bags. The observations and videos showed no issues with the back (roof of the mine). No 
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